Author's response to reviews

Title: Dental sealant knowledge, opinion, values and practice of Spanish dentists

Authors:

Laura San Martin (LSanMartin@us.es)
Antonio Castaño (acastano@us.es)
Manuel Bravo (mbravo@ugr.es)
Mary Tavares (mtavares@forsyth.org)
Richard Niederman (rniederman@forsyth.org)
Eyitope O Ogunbodede (Eyitope_Ogunbodede@hsdm.harvard.edu)

Version: 2 Date: 14 September 2012

Author's response to reviews: see over
Dear Editor,

Article title: Dental sealant knowledge, opinion, values and practice of Spanish dentists
MS ID: 4180603417963638
Authors: Laura San Martin, Antonio Castaño, Manuel Bravo, Mary Tavares, Richard Niederman and Eyitope O Ogunbodede
Journal: BMC Oral Health

Please refer to your email of 2-12-09-03 on the above subject matter. We have revised the manuscript as suggested and the following changes have been made:

(1) On the issue of ethical approval, a statement has been added to the Methods section of the manuscript - “Approval for the study was obtained from Comisión de Doctorado de la Universidad de Sevilla, Spain.” (Last statement of Paragraph 2 on Page 7). A copy of the

(2) Informed consent: Since this is a questionnaire study, the purpose of the study was explained to participants and acceptance to complete the questionnaire was taken as consent to the study.

(3) The last sentence of the acknowledgement has been deleted.

We hope with the above clarifications, the manuscript can now be sent for peer review.

Thank you.

Regards,

Dr. San Martin